# FUNCTIONAL OUTCOMES OF REHABILITATION IN PATIENTS WITH SUBACUTE STROKE ON HAEMODIALYSIS

Goro SHINJO, MD<sup>1</sup>, Yohei OTAKA<sup>®</sup>, MD, PhD<sup>2,3</sup>, Kaori MURAOKA, MD, MPH, PhD<sup>2</sup>, Hiroki EBATA, MD, PhD<sup>1</sup> and Meigen LIU, MD, PhD<sup>2</sup>

From the <sup>1</sup>Department of Rehabilitation Medicine, Saiseikai Kanagawaken Hospital, Kanagawa, <sup>2</sup>Department of Rehabilitation Medicine, Keio University School of Medicine, Tokyo and <sup>3</sup>Department of Rehabilitation Medicine I, Fujita Health University School of Medicine, Aichi, Japan

*Objective:* To examine functional outcomes in patients with subacute stroke who are on haemodialysis.

Design: Observational cohort study.

Subjects: Patients with stroke who were admitted to rehabilitation wards between April 2011 and August 2016 were retrospectively enrolled.

Methods: Data on the Functional Independence Measure (FIM) and discharge destination were collected from medical records as outcomes. Outcomes were compared for patients who received haemodialysis and those who did not. Multiple regression analyses were carried out to explore the impact of haemodialysis on functional outcomes.

**Results:** Among 859 patients, 34 (mean age 63.6 years (standard deviation (SD) 11.7)) required haemodialysis while staying in rehabilitation wards. One hundred patients not receiving haemodialysis (mean age 70.7 years (SD 13.8)) were randomly selected as controls. The total FIM score at discharge, FIM gain, FIM efficiency, FIM effectiveness, and discharge destination were not significantly different between the 2 groups (p > 0.05). Multiple regression analysis, adjusted for other confounding factors, found no significant difference in total FIM score between groups at discharge (p = 0.427).

*Conclusion:* Dialysis was not an inhibitory factor affecting outcomes during subacute rehabilitation in patients with stroke. Rehabilitation should be considered for patients with subacute stroke requiring haemodialysis, in order to provide every opportunity to achieve adequate functional outcomes.

Key words: activities of daily living; comorbidity; functional outcome; kidney diseases; rehabilitation; renal dialysis; stroke.

Accepted Jan 13, 2020; Epub ahead of print Jan 24, 2020

J Rehabil Med 2020; 52: jrm00033

Correspondence address: Yohei Otaka, Department of Rehabilitation Medicine I, School of Medicine, Fujita Health University, 1-98 Dengakugakubo, Kutsukake, Toyoake, Aichi, 470-1192, Japan. E-mail: otaka119@mac.com

Chronic kidney disease (CKD), including end-stage renal failure, is associated with platelet dysfunction, coagulation disorder, endothelial dysfunction, inflammation, alterations in cardiac output and regional cerebral perfusion, accelerated systemic atherosclero-

## LAY ABSTRACT

The aim of this study was to clarify whether patients with subacute stroke who are receiving haemodialysis can attain similar functional outcomes after rehabilitation as those not receiving haemodialysis. Stroke patients who were admitted to rehabilitation wards were retrospectively enrolled. Functional outcomes were compared between 34 patients receiving haemodialysis and 100 randomly selected patients not receiving haemodialysis. The results indicate that the functional outcomes were not significantly different between the groups. Furthermore, dialysis was proven to not be an inhibitory factor that affects outcomes after controlling for possible cofounding factors. This study suggests that clinicians should consider rehabilitation for all patients with subacute stroke who require haemodialysis, in order to provide the opportunity for these patients to achieve adequate functional outcomes.

sis, altered blood brain barrier, and the risk of atrial fibrillation; thus, CKD is a risk factor for cerebrovascular diseases (1–5). The incidence rate of ischaemic stroke and haemorrhagic stroke in patients with CKD is much higher than in the general population (6, 7). In addition, the number of patients with end-stage renal disease requiring haemodialysis (HD) is increasing in developed countries, including Japan (8–10). Therefore, clinicians are more frequently encountering stroke patients on HD who are candidates for rehabilitation.

It is well-established that rehabilitation outcomes in general deteriorate with an increasing number of comorbidities and severity of comorbidities (11–13). Patients with end-stage renal failure are more likely to be restricted in daily activities because of various unfavourable conditions, including anaemia, decreased cardiac function, protein energy malnutrition, lower muscle strength, and metabolic disorders (14, 15). However, the effect of CKD on functional outcomes in stroke patients is not well understood. One study (16) with a limited sample of patients with cerebrovascular disease (n=11) reported that HD patients with cerebrovascular disease had lower motor score gains on the Functional Independence Measure (FIM) than did non-HD patients. The lack of knowledge about functional outcomes in stroke patients with CKD inhibits appropriate decision-making, particularly

regarding whether an intensive inpatient rehabilitation programme is relevant to a stroke patient on HD.

The aims of this retrospective study were: (i) to investigate the characteristics of patients with subacute stroke on HD receiving rehabilitation; and (ii) to explore whether the rehabilitation outcomes were different between subacute stroke patients receiving HD and those who were not receiving HD.

## **METHODS**

### Patients

JRM

A total of 859 consecutive patients with cerebral infarction or cerebral haemorrhage (excluding subarachnoid haemorrhage) who were transferred to the Kaifukuki Rehabilitation Ward (KRW) in Saiseikai Kanagawaken Hospital from acute hospitals for rehabilitation between April 2011 and August 2016, were reviewed retrospectively. Saiseikai Kanagawaken Hospital is the only rehabilitation facility in Kanagawa-ku at which HD can be performed. Kanagawa-ku is 1 of the 18 wards of the city of Yokohama in Kanagawa Prefecture, Japan. It has an estimated population of 238,578, covering an area of 31.01 km<sup>2</sup>. Before the review, we estimated the sample size for the analyses. We determined that a sample size of 29 patients for the HD group and 3 times that for the non-HD group (n=87) was sufficient to detect a difference of 9 points in the FIM score, assuming a standard deviation (SD) of 16 using a 2-tailed t-test of difference between means, with 80% power and a 5% level of significance. For the collection of patients in the HD group, we started with patients seen in August 2016 and worked backwards in time until the sample size was reached. Thirty-four patients with HD were then enrolled, including patients seen between April 2011 and August 2016. For selection of non-HD patients, 100 patients were randomly selected by computer-generated random numbers from a group of patients admitted during the same period who did not receive HD. This study was approved by the ethics committee of Saiseikai Kanagawaken Hospital (No. 16-09).

#### Subacute rehabilitation in Japan

The KRW system for intensive rehabilitation, a governmental insurance system for rehabilitation of patients in rehabilitation wards during their convalescent period, was introduced in Japan in 2000. Patients were eligible for admission to the KRW within 2 months of onset of stroke. Physical, occupational, and speech therapies lasting for a maximum of 3 h/day were provided as part of the intensive intervention. The KRW team provided patients and their families with a comprehensive monthly rehabilitation plan, including information about achieved goals, planned goals, and rehabilitative approaches to the achieved goals, discharge planning, and social resources necessary for home discharge. Timing for discharge was set when patients reached a plateau in activities of daily living, and the maximum length of stay was 150 days for patients with stroke and 180 days for patients with stroke and accompanying severe disabilities and cognitive disorders.

#### Data collection

Patient information, including age, sex, days from onset to admission, past history of stroke, types of stroke, side of hemiparesis, motor function on the Stroke Impairment Assessment Set (17), Charlson Comorbidity Index (18), transient discontinuation of rehabilitation, length of stay in the rehabilitation hospital, discharge destination, modified Rankin scale (mRS) (19) before the onset and at discharge, and admission and discharge FIM (20, 21), was collected. Transient discontinuation of rehabilitation was defined as an event of discontinued rehabilitation for medical reasons lasting one day or more. In addition, for the patients on HD, data about the type of renal disease and the duration of dialysis were collected. Information on the total time of rehabilitation undertaken per day was collected and defined as the dose of rehabilitation.

#### Analyses

Prior to analyses between the HD and non-HD groups, we examined whether the selected non-HD group was a representative sample of whole patients who were not receiving HD. Age, sex, and stroke types were compared between the selected non-HD patients and the rest of the patients who were not receiving HD using an independent *t*-test or  $\chi^2$  test.

For baseline comparisons, demographic and clinical variables at admission were compared in the HD and non-HD groups using the independent *t*-test, Mann–Whitney test, or  $\chi^2$  test, depending on the type of variables.

Discharge FIM, FIM gain, FIM efficiency, and FIM effectiveness were employed as the functional outcomes (22). Indices were calculated using the following equations:

Gain=discharge FIM-admission FIM

Efficiency=(discharge FIM-admission FIM)/Length of stay Effectiveness=(discharge FIM-admission FIM)/(maximum FIM-admission FIM)

Length of stay in the rehabilitation hospital, mean rehabilitation time per day, the number of patients who experienced transient discontinuation of rehabilitation, mRS at discharge, and discharge destination were used as the other outcomes. Outcomes were compared between groups with the Mann–Whitney test or  $\chi^2$  test, depending on the types of variables.

To investigate whether there was a significant difference in the discharge FIM between the HD group and the non-HD group after adjusting for covariates and possibly confounding variables, we developed a multiple regression model for the discharge FIM with group (HD/non-HD), age, sex, length of hospital stay, admission FIM, mean rehabilitation time per day, and the Charlson Comorbidity Index as covariates. JMP software version 13 (SAS Institute Inc., Cary, NC, USA) was used for statistical analyses. Any *p*-value <0.05 was considered statistically significant with no adjustment for multiple comparisons.

## RESULTS

The flow diagram of patients is shown in Fig. 1. There was no significant difference in age (mean 70.7 (SD 13.8) vs 71.1 (SD 12.5) years, p=0.796), sex (male/female, 65/35 vs 425/300, p=0.280), or types of diseases (haemorrhage/infarction, 44/56 vs 248/477, p=0.116) in randomly selected non-HD patients (n=100) and the rest of the non-HD patients (n=725).

The characteristics of the patients on admission to the rehabilitation ward in each group are shown in Table I. The causes of HD were diabetes (n=18), nephrosclerosis (n=6), glomerulonephritis (n=4), polycystic kidney (n=1), and unknown cause (n=5). The mean duration after the introduction of HD was







Fig. 1. Flow diagram of patients admitted to Saiseikai Kanagawaken Hospital with cerebral infarction or cerebral haemorrhage between April 2011 and August 2016. HD: haemodialysis.

8.1 years (SD 6.6). Compared with the non-HD group, patients in the HD group were significantly younger, had a higher Charlson Comorbidity Index, milder upper limb paresis, and better motor, cognitive, and total FIM scores at admission (Table I).

A patient in the HD group died in the KRW due to severe aspiration pneumonia. The data of this case has been included in the analyses. There were no significant differences in total and motor FIM scores at discharge, FIM gain, FIM efficiency, FIM effectiveness, the number of patients with transient discontinuation of rehabilitation, mRS at discharge, and discharge destination between groups (Tables II and III). The HD

**Table I.** Characteristics of participants on admission to rehabilitation wards

	HD group (n=34)	Non-HD group ( <i>n</i> =100)	<i>p</i> -value
Age, years, mean (SD)	63.6 (11.7)	70.7 (13.8)	0.008
Sex, male/female, n	21/13	65/35	0.736
Type of stroke, haemorrhage/infarction, n	10/24	44/56	0.152
mRS before onset, median (IQR)	0 (3)	0 (2)	0.975
Charlson Comorbidity Index, median (IQR)	6 (2)	2 (2)	< 0.001
Past history of stroke, n (%)	11 (32)	21 (21)	0.180
Duration from onset to admission, days, mean (SD)	34.7 (21.6)	36.5 (14.6)	0.596
SIAS at admission, median (IQR)			
Knee mouth test	4 (3)	3 (4)	0.030
Finger function test	4 (2.25)	3 (4)	0.040
Hip flexion test	4 (2)	3.5 (3)	0.399
Knee extension test	3.5 (2.5)	3 (3)	0.208
Foot pad test	3.5 (3)	3 (4)	0.329
Total	17.5 (12)	15 (18)	0.156
FIM at admission, median (IQR)			
Motor score	59 (39.5)	40 (42.5)	0.035
Cognitive score	29 (26)	20.5 (15.5)	0.017
Total score	86.5 (51.5)	60 (58.5)	0.022

IQR: interquartile range; HD: haemodialysis; mRS: modified Rankin scale; SIAS: Stroke Impairment Assessment Set; FIM: Functional Independence Measure; SD: standard deviation.

Table II. Comparison	of outcomes	of Functional	Independence
Measure between group	S		

	HD group ( <i>n</i> =34) Median (IQR)	Non-HD group (n=100) Median (IQR)	<i>p</i> -value
Motor score at discharge	77 (42)	63 (46.5)	0.229
Cognitive score at discharge	29.5 (12)	27 (14)	0.047
Total score at discharge	107 (54)	89.5 (62.5)	0.094
Gain	11.5 (14)	16 (20)	0.089
Efficiency	0.16 (0.17)	0.15 (0.15)	0.925
Effectiveness	0.34 (0.47)	0.40 (0.47)	0.864

IQR: interquartile range; HD: haemodialysis.

group showed significantly shorter lengths of hospital stay. In addition, the HD group showed significantly higher cognitive FIM scores at discharge, although this might be mainly attributed to the significant group difference in the scores at admission. The mean time of rehabilitation per day was 25% shorter in patients on HD than in non-HD patients (Table III). Reasons for the transient discontinuation of rehabilitation were shunt stenosis (n=2), renal anaemia (n=2), heart failure (n=2), and "other" (n=4) in the HD group. In the non-HD group, the reasons were aspiration pneumonia (n=6), epilepsy (n=3), cerebral infarction (n=1), chronic subdural haematoma (n=1), and "other" (n=9).

Multiple regression analysis revealed that the group (HD over non-HD) was not significant. The factors related to the discharge FIM score were the length

Table III. C	Comparison	of other	outcomes	between	groups
--------------	------------	----------	----------	---------	--------

•			
	HD group ( <i>n</i> =34)	Non-HD group (n=100)	<i>p</i> -value
Length of hospital stay, days, mean (SD)	82.7 (50.3)	119.4 (50.3)	< 0.001
Dose of rehabilitation/day, min, mean (SD)	99.9 (28.5)	132.9 (17.0)	< 0.001
Patients with transient discontinuation of rehabilitation, $n$ (%)	11 (32)	20 (20)	0.140
mRS at discharge, median (IQR)	3 (2)	4 (1)	0.673
Discharge destination, home/others, n	23/11	70/30	0.797

IQR: interquartile range; HD: haemodialysis; mRS: modified Rankin Scale; SD: standard deviation.

 $\label{eq:table_to_stability} \begin{array}{l} \textbf{Table IV.} \\ \textbf{Multiple regression analysis for Functional Independence} \\ \textbf{Measure (FIM) score at discharge} \end{array}$ 

	Unstandardized coefficient	Standard error	Standardized coefficient	t	<i>p</i> -value
Group, HD over					
non-HD	-1.76	2.20	-0.05	-0.80	0.427
Age, year	-0.15	0.10	-0.06	-1.54	0.126
Sex, female	0.98	1.29	0.03	0.75	0.453
Length of hospital stay	0.09	0.03	0.14	2.93	0.004
Admission total FIM score	1.01	0.05	0.96	21.80	< 0.001
Dose of rehabilitation per day, min	0.20	0.06	0.15	3.13	0.002
Charlson Comorbidity					
Index	-0.19	1.01	-0.01	-0.19	0.848
Intercept	-6.53	12.74	0.00	-0.51	0.609

Adjusted  $R^2 = 0.83$ . F(7,126) = 95.076, p < 0.001, R = 0.92.

HD: haemodialysis.

of hospital stay, total admission FIM score, and the dose of rehabilitation per day (Table IV). Even after excluding the Charlson Comorbidity Index from the model, which also added points for moderate to severe CKD, the result still showed that the group was not significant.

## DISCUSSION

This study investigated the characteristics of stroke patients on HD who were receiving subacute rehabilitation, and explored whether the rehabilitation outcomes were different in patients on HD and non-HD patients with subacute stroke in a subacute rehabilitation ward. The patients on HD were significantly younger and had significantly milder upper limb paresis and significantly higher FIM score than non-HD patients at baseline. On the comparison of outcomes between groups, discharge total FIM score, FIM gain, FIM efficiency, FIM effectiveness, and discharge destination were not significantly different between groups (p > 0.05), even though the mean dose of rehabilitation per day was 25% shorter in patients on HD.

Some studies examined the outcomes of stroke patients on HD. Mattana et al. (23) reported that stroke patients on HD could have as good an outcome as those with normal renal function, but were hospitalized longer in an acute hospital setting. Sozio et al. (24) revealed in a large cohort that only 56% of patients successfully escaped death, nursing home, or skilled nursing facility. To the best of our knowledge, however, the characteristics and outcomes of patients on HD receiving rehabilitation have been investigated in only 2 studies with limited numbers of stroke samples, one with only 2 patients (25) and one with only 11 stroke patients (16). Therefore, our findings have valuable implications, even though the sample was not large.

With regard to baseline characteristics, we revealed that patients on HD were in significantly better condition than non-HD patients at baseline. Selection bias regarding candidacy for rehabilitation is the most likely reason for this baseline difference between groups. Although there was no definite evidence that patients on HD will have poorer outcomes compared with non-HD patients, clinicians may tend to choose non-HD patients rather than patients on HD when other conditions are similar, because there are a limited number of rehabilitation hospitals where HD can be performed.

Regarding the rehabilitation outcomes, Forrest et al. (25) reported that the rate of patient discharge to the community was almost the same in patients on HD (34 patients, including 2 stroke) and non-HD patients (497 patients, including 44 stroke). The mean rehabilitation

time per day was 27% shorter in patients on HD than non-HD patients. These findings were similar to ours. However, they did not report the findings limited to stroke patients and only 2 HD stroke patients were included. Therefore, these findings cannot be compared with our results. Cowen et al. (16) reported that gains in motor FIM during hospitalization were significantly lower in 11 HD stroke patients with end-stage renal disease compared with those without it. Although the finding also cannot be simply compared with ours because of the limited number of samples and a much shorter mean length of stay in hospital (<20 days), the finding was different from ours, where FIM gain, efficiency, and effectiveness were not different in HD and non-HD patients. There are 2 possible interpretations for our finding. First, different baseline characteristics might have contributed to the findings. The significantly better condition of HD patients at baseline may have cancelled out the negative effects resulting from intrinsic negative factors in patients on HD, as well as the negative influence of a lesser dose of rehabilitation. The other possible interpretation is that the effects of rehabilitation are truly similar between patients on HD and non-HD patients when rehabilitation is provided over a relatively long-term period in the subacute phase. This explanation was supported by the finding that the discharge total FIM scores were not significantly different between the HD group and non-HD group even after accounting for age, admission FIM, and other possible confounding variables in the multiple regression analysis. We could not conclude which of these interpretations is true: however, we can at least draw the clinical conclusion that HD patients may have a comparable rehabilitation effect to that of non-HD patients during the subacute period.

This study had several potential limitations. The KRW, where the study was conducted, is a unique system for subacute rehabilitation in Japan; and the study was conducted in a single facility. The design of the study was retrospective. Therefore, the generalizability of the findings is limited. Furthermore, although we tried to eliminate the effects of the difference in characteristics between the groups by using a multiple regression model, there is the possibility that the influence of this baseline difference could not be completely removed.

In conclusion, patients on HD with subacute stroke in the rehabilitation ward could have had favourable rehabilitation outcomes, which were comparable to those in non-HD patients. The findings of this study suggest that HD is not a major inhibitory factor for rehabilitation and patients on HD with stroke should have the same opportunities as non-HD stroke patients, because they may benefit from rehabilitation. The authors have no conflicts of interest to declare.

## REFERENCES

- El Husseini N, Kaskar O, Goldstein LB. Chronic kidney disease and stroke. Adv Chronic Kidney Dis 2014; 21: 500–508.
- Lee M, Saver JL, Chang KH, Liao HW, Chang SC, Ovbiagele B. Impact of microalbuminuria on incident stroke: a metaanalysis. Stroke 2010; 41: 2625–2631.
- Masson P, Kelly PJ, Craig JC, Lindley RI, Webster AC. Risk of stroke in patients with ESRD. Clin J Am Soc Nephrol 2015; 10: 1585–1592.
- Toyoda K, Ninomiya T. Stroke and cerebrovascular diseases in patients with chronic kidney disease. Lancet Neurol 2014; 13: 823–833.
- Ghoshal S, Freedman BI. Mechanisms of stroke in patients with chronic kidney disease. Am J Nephrol 2019; 50: 229–239.
- Wang HH, Hung SY, Sung JM, Hung KY, Wang JD. Risk of stroke in long-term dialysis patients compared with the general population. Am J Kidney Dis 2014; 63: 604–611.
- 7. Alqahtani F, Berzingi CO, Aljohani S, Al Hajji, Diab A, Alvi M, et al. Temporal trends in the outcomes of dialysis patients admitted with acute ischemic stroke. J Am Heart Assoc 2018; 7: e008686.
- Foley RN, Collins AJ. End-stage renal disease in the United States: an update from the United States Renal Data System. J Am Soc Nephrol 2007; 18: 2644–2648.
- 9. Lee G. End-stage renal disease in the Asian-Pacific region. Semin Nephrol 2003; 23: 107–114.
- 10. Prasad N, Jha V. Hemodialysis in Asia. Kidney Dis (Basel) 2015; 1: 165–177.
- Berlowitz DR, Hoenig H, Cowper DC, Duncan PW, Vogel WB. Impact of comorbidities on stroke rehabilitation outcomes: does the method matter? Arch Phys Med Rehabil 2008; 89: 1903–1906.
- Liu M, Domen K, Chino N. Comorbidity measures for stroke outcome research: a preliminary study. Arch Phys Med Rehabil 1997; 78: 166–172.
- Liu M, Tsuji T, Tsujiuchi K, Chino N. Comorbidities in stroke patients as assessed with a newly developed comorbidity

scale. Am J Phys Med Rehabil 1999; 78: 416-424.

- Deligiannis A, Kouidi E, Tassoulas E, Gigis P, Tourkantonis A, Coats A. Cardiac effects of exercise rehabilitation in hemodialysis patients. Int J Cardiol 1999; 70: 253–266.
- Kouidi E, Albani M, Natsis K, Megalopoulos A, Gigis P, Guiba-Tziampiri O, et al. The effects of exercise training on muscle atrophy in haemodialysis patients. Nephrol Dial Transplant 1998; 13: 685–699.
- Cowen TD, Huang CT, Lebow J, DeVivo MJ, Hawkins LN. Functional outcomes after inpatient rehabilitation of patients with end-stage renal disease. Arch Phys Med Rehabil 1995; 76: 355–359.
- Chino N, Sonoda S, Domen K, Saitoh E, Kimura A. Stroke Impairment Assessment Set (SIAS): a new evaluation instrument for stroke patients. Jpn J Rehabil Med 1994; 31: 119–125.
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis 1987; 40: 373–383.
- van Swieten JC, Koudstaal PJ, Visser MC, Schouten HJ, van Gijn J. Interobserver agreement for the assessment of handicap in stroke patients. Stroke 1988; 19: 604–607.
- 20. Keith RA, Granger CV, Hamilton BB, Sherwin FS. The functional independence measure: a new tool for rehabilitation. Adv Clin Rehab 1987; 1: 6–18.
- 21. The Data Management Service of the Uniform Data System for Medical Rehabilitation and the Center for Functional Assessment Research: guide for use of the uniform data set for medical rehabilitation (Ver. 3.0). New York: State University of New York at Buffalo; 1990.
- Koh GC, Chen CH, Petrella R, Thind A. Rehabilitation impact indices and their independent predictors: a systematic review. BMJ Open 2013; 3: e003483.
- Mattana J, Effiong C, Gooneratne R, Singhal PC. Outcome of stroke in patients undergoing hemodialysis. Arch Intern Med 1998; 158: 537–541.
- 24. Sozio SM, Armstrong PA, Coresh J, Jaar BG, Fink NE, Plantinga LC, et al. Cerebrovascular disease incidence, characteristics, and outcomes in patients initiating dialysis: the choices for healthy outcomes in caring for ESRD (CHOICE) study. Am J Kidney Dis 2009; 54: 468–477.
- 25. Forrest GP. Inpatient rehabilitation of patients requiring hemodialysis. Arch Phys Med Rehabil 2004; 85: 51–53.